Author's response to reviews

Title: Drug therapy for hereditary cancers

Authors:

Evgeny N Imyanitov (evgeny@imyanitov.spb.ru)
Vladimir M Moiseyenko (moiseyenko@gmail.com)

Version: 2 Date: 22 July 2011

Author's response to reviews:

Dear Editors,

We appreciate greatly your positive consideration of this manuscript.

Please find below the response to the suggestions of the reviewers.

Referee #2:

Comment: The last paragraph of page 8 requires some attention as the authors have not described any details about the patient’s disease. I presume it is pancreatic cancer but am not certain.
Response: The diagnosis of the pancreatic cancer has been specified in the text.

Comment: The authors should describe the potential value of aspirin in HNPCC.
Response: We incorporated an appropriate paragraph in the end of the “Hereditary non-polyposis colorectal cancer” section.

We hope that the manuscript is now acceptable for the publication and look forward to the decision of the Editorial Board.

Best wishes

Sincerely

Evgeny Imyanitov